Author’s response to reviews

Title: Cross-cultural validation of the Bengali version Kidscreen-27 quality of life questionnaire

Authors:

Rosalie Power (rosalie.power@gmail.com)
Rahena Akhter (rahena.akhter@sydney.edu.au)
Mohammad Muhit (mmuhit@hotmail.com)
Sabrina Wadud (swadud12@yahoo.com)
Eamin Heanoy (eamin7heanoy@gmail.com)
Tasneem Karim (tasneem.karim.tk@gmail.com)
Nadia Badawi (nadia.badawi@health.nsw.gov.au)
Gulam Khandaker (Gulam.Khandaker@health.qld.gov.au)

Version: 1 Date: 06 Jul 2018

Author’s response to reviews:

Dear Editor,

Thank you for your feedback. Requested changes to the manuscript have been implemented including

- Consent to participate
- Consent to publish
- Abbreviations
- Table citation
- Authors email
- Authors contribution

- Clean manuscript

Kind regards,

Rosalie (on behalf of authorship team)